Author's response to reviews

Title: Protocol of a prospective cohort study of the effect of different methods of drainage of neuropathic bladder on occurrence of symptomatic urinary infection, and adverse events related to the urinary drainage system in spinal cord injury patients

Authors:

Subramanian Vaidyanathan (vaidyanathansiu@hotmail.com)
Bhakul M Soni (Bhakul.Soni@mail.soh-tr.nwest.nhs.uk)
Singh Gurpreet (Gurpreet.Singh@mail.soh-tr.nwest.nhs.uk)
Paul Mansour (Paul.Mansour@mail.soh-tr.nwest.nhs.uk)
Peter L Hughes (Peter.Hughes@mail.soh-tr.nwest.nhs.uk)
Tun Oo (drtunoo@hotmail.com)
Pradipkumar Sett (Pradip.Sett@mail.soh-tr.nwest.nhs.uk)
Keith F Parsons (Keith_Parsons@hotmail.com)
John C Davies (J.C.Davies@liverpool.ac.uk)

Version: 7 Date: 20 Nov 2001

The manuscript has been revised as per the advice of Dr Carole Mongin-Bulewski, PhD, Editorial Assistant, BioMedCentral.